Author’s response to reviews

Title: Socio-demographic determinants of coinfections by HIV, hepatitis B and hepatitis C viruses in central Italian prisoners

Authors:

Giuseppe La Torre (giuseppe.latorre@rm.unicatt.it)
Luca Miele (luca.miele@rm.unicatt.it)
Giacomina Chiaradia (gemmachiaradia@yahoo.it)
Alice Mannocci (alice.mannocci@rm.unicatt.it)
Manuela Reali (manuelareali@hotmail.com)
Giovanni Gasbarrini (ggasbarrini@rm.unicatt.it)
Elisabetta De Vito (devito@unicas.it)
Antonio Grieco (agrieco@rm.unicatt.it)
Walter Ricciardi (wricciardi@rm.unicatt.it)

Version: 4 Date: 21 August 2007

Author’s response to reviews:

Cover letter MS 1949767766127357
Revision 18-8-07

We made the following changes:

1. we wrote each affiliation in full
2. we inserted the authors’ contribution
3. we checked the references and made changes on the ref. number 50 and 53, reporting only the first six authors and then et al.
4. we checked the reference citation in the text, and we used a comma instead of a hyphen when citing two consecutive reference numbers.
5. we removed the full stop from the end of the tables’ title
6. we cited the tables in the text in the same way, i.e. Table